\begin{table}%t2
\caption{\label{t:parameters}Stellar parameters of the CoRoT exoplanets parent stars.}
%%\centerline

{\small
\begin{tabular}{lccccccc}
\hline
\hline\noalign{\smallskip}
Star Name & $T_{\rm eff}  $[K]  & $\log~ g$  &  $\xi_{\rm t}$ [km~s$^{-1}$] & [Fe/H] &  [M/H] &  $v \sin i$ [km~s$^{-1}$]  & References   \\
\hline
\\
\object{CoRoT-2 }  & 5696 $\pm$ 70 & 4.42 $\pm$ 0.12 & 1.71 $\pm$ 0.10  &  0.03  $\pm$ 0.06    & 0.03 $\pm$ 0.06 & 8.5 $\pm$ 1 & This work   \\
\object{CoRoT-4 }  & 6115 $\pm$ 70 & 4.30 $\pm$ 0.12 & 1.37  $\pm$ 0.10 &   0.12  $\pm$  0.05  & 0.10 $\pm$ 0.05 &  5.5 $\pm$ 1  & This work    \\
\hline
\\
CoRoT-2           & 5625 $\pm$ 120 & 4.30 $\pm$ 0.20 &  --             & --               & 0.0 $\pm$ 0.1 & 11.85 $\pm$ 0.45  &   \cite{bouchy08} \\
                  & 5608 $\pm$ 37  & 4.71 $\pm$ 0.20 & 1.49 $\pm$ 0.06 & 0.07 $\pm$ 0.04  &    --           &     --              &   \cite{ammlervon09}\\
CoRoT-4           & 6190 $\pm$ 60 & 4.41 $\pm$ 0.05 & 0.94  $\pm$ 0.05 &   --  & 0.05 $\pm$ 0.07 & 6.4 $\pm$ 1 & \cite{moutou08}     \\
\object{CoRoT-3 } & 6740 $\pm$ 140  & 4.22 $\pm$ 0.07   & -- & 0.03  $\pm$ 0.06  & $-0.02 \pm 0.06$ & 17 $\pm$ 1 &\cite{deleuil08}   \\
\object{CoRoT-5 } & 6100 $\pm$  65  & 4.189 $\pm$ 0.03  & 0.91 $\pm$ 0.09 &  $-0.25 \pm 0.06$ & $-0.25 \pm 0.06$ & 1 $\pm$ 1 & \cite{rauer09}\\
\hline
\end{tabular}}
\tablefoot{Column~1: star name; Col.~2: effective temperature; Col.~3: surface gravity; Col.~4: microturbulence; Col.~5: metallicity; Col.~6: mean abundance (using element with  more than three lines);  Col.~7: projected rotational velocity; Col.~8: references.}
\end{table}